\begin{table}%T1
\par
      \caption{Published ISO SWS spectra of carbon stars with an SiC feature.
      The third column lists the features published in the literature listed in the last column.}
         \label{table 1}
$$
         \begin{array}{llll}
            \hline\hline
            \noalign{\smallskip}
            IRAS &  Name    &   {\it Spectral}~ features ~(unit{:}~ \mu{\rm m})     &  References  \\
           \noalign{\smallskip}
            \hline%
            \noalign{\smallskip}
01246-3248  & R\hspace{0.1cm}Scl       & 2.48,2.58,3.05,3.9,5.20,11.30,13.70          &  (1), (2),(3) \\
02293+5748  & AFGL\hspace{0.1cm}341       & 11.30, 13.7 & (4) \\
12226+0102  & SS\hspace{0.1cm}Vir      & 2.48,2.58,3.05 ,5.20,11.30,13.7              &  (4)\\
12447+0425  & RU\hspace{0.1cm}Vir      & 2.48,2.58,3.05,3.9,5.20,11.30,13.70          &  (2)\\
15477+3943  & V\hspace{0.1cm}CrB       & 3.05,11.3,13.7                               &  (4)\\
17556+5813  & T\hspace{0.1cm}Dra       & 3.05,11.3,13.7                               &  (5), (6)\\
19017-0545  & V\hspace{0.1cm}Aql       & 2.48,2.58,3.05,5.20,11.30,13.70              &  (1)\\
20396+4757  & V\hspace{0.1cm}Cyg       & 2.48,2.58,3.05,3.9,5.20,11.30,13.70          &  (1)\\
21399+3516  & V460\hspace{0.1cm}Cyg    & 3.50,5.20,11.3                               &  (7),(8) \\
21489+5301  &  $/$        & 11.30,13.70                                  &  (9)\\
22036+3315  & RZ\hspace{0.1cm}Peg      & 3.50,3.56,3.9,11.30                          &  (7)\\
23320+4316  & LP\hspace{0.1cm}And      & 3.05,11.30,13.70                             &  (1)\\
            \hline
         \end{array}
$$
%\begin{flushleft}
  (1) Yamamura et al. (\cite{Yama}); (2) Hron et al. (\cite{Hronb});
(3) Hron et al. (\cite{Hronc}); (4) Aoki et al. (\cite{Aokb});
  (5) Hron et al. (\cite{Hrona}); (6) Loidl et al. (\cite{Loidb});
 (7) Aoki et al. (\cite{Aoka}); (8) J$\o$rgensen et al. (\cite{Jrge});
   (9) Volk et al. (\cite{Volkb}).
% \end{flushleft}
 \end{table}